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0730 Red Coal Creek Fire Department cancelled playing in exercise as of 0730 am
0745 Red Red net radio checks complete
0800 Red current weather passed to control organizations
0825 Green Green net radio check complete-green 9 in place @ OMD
0827 Green Contacted Green net to inform exercise inititation in approx 3 minutes
0829 Yellow Radio checks complete - all in position and ready

0840 Yellow
Exercise on hold due to surveillance in B371/374 (combustable control surveillance) initiation 
approx 0900 - yellow net informed

0845 Red
per red 1, controller was ejected from EOC.  He had unauthorized electronic equipment on his 
person.  Will be allowed access after electronic equipment is removed.

0845 Red
Broomfield Ambulance is having difficulty gaining access to Site.  Monte Wright is working 
issue but currently remains unresolved. 

0847 Red Broomfield ambulance is on-Site.  Issue resolved.
0850 Red EOC Controller corrected earlier issue and has returned to EOC
0850 Yellow Resetting scene
0852 Red Weather update
0855 Green contacted Green - weather update and exercise initiation
0857 Blue Event initated @ B374
0857 Green contacted Green net - exercise initiated @ 0857
0857 Yellow 1 Exercise initiated - yellow net informed

0857-E ECCM 1 Exercise Initiation
0859 Yellow 2 Yellow 1 reporting message #2 dropped
0900 Blue Fire Dept units clearing PACS 2
0900 White 7 Fire Dept units thru Portal 2 to 374 fire
0900 Yellow LS/DW announcement in B371/374 to perform controlled evacuation
0901 Blue Fire Dept units placement on N & E sides of Bldg 374
0901 White LIMA 2 confirm - no vaults in access
0902 White Pro Force escorting Fire Dept thru Portal 2

0902 Yellow
Yellow 4 reporting worker #6 cannot find anyone from building in charge of accountability - 
instructed by yellow control to play out - facility management should be arriving

0903 White 6 Fire Dept units in PA
0903 White TCP’s deploying
0904 White Pro Force to EOC for access control
0906 White Fire Department moving all personnel NE of B374
0907 Green green 9 reporting LS/DW @ B122 & 125 outside not working

0908 Red
604 has left EOC, declared an operational emergency at ALERT*, directed activation of EOC, 
ERO and performance of offsite notifications

0908 White LIMA 2 closing portal 2
0908 Yellow Yellow-4 being asked if accountability of facility personnel has started.  Yes as of 0908
0908 Yellow 8 asked Yellow-4 the time message #8 was dropped (0908)
0909 Blue Fire Department preparing to enter structure
0910 Blue ICP established by Fire Dept to the N & E of B374; 2-100,000 gal tanks @ B374
0910 Green Green 1 reported fire hosess are charged.  Instructed G-1 to STOP fire hose charge
0910 Green Green-1 reporting no fire controller in NE section
0910 White TOC activated
0910 White LIMA 2 moving personnel to secondary assembly area
0911 Blue victims #4 & #5 contacted by Fire Dept personnel

0913 Blue
LS/DW announcement : 1-B374 personnel evacuate to NE corner of builidng; 2-ERO personnel 
to ICP; 3-safe route of travel

0913 Received via fax to 5206 Emergency Offsite Notification Form-initial
0914 Blue ERO page received by Blue control
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0914 Blue Advised Blue-8 that Fire Dept is not to used charged hose lines
0915 Blue Blue-7 reporting simulated hoselines for building interior have been approved

0915 Green
Hot zone set up by Fire Dept on the N side of 374.  1 victim removed & first aid is being 
administered

0915 Yellow Yellow-1 reporting Fire Dept has discharged fire hoses and have entered the facility
0916 Blue 4 Messge #4 dropped by Blue-7
0916 Blue Fire Dept EMS personnel treating victims #4 & #5 on exterior of building, NE corner
0916 Green told that message #3 was not dropped- Fire Dept entered bldg first with no RCT/s
0919 Blue S/S on-scene @ Fire Dept ICP
0920 Blue 5 Message #5 dropped by Blue-7
0920 Green Green-4 stated that victims #4 & #5 were removed from southside to northside of B374
0920 White Unit 240 within affected area contaminated by RCT
0922 Green 4 message #4 dropped by Green-2

0922 Yellow
Yellow-4 reporting no RAD Ops controller at accountability board-directions?  Y-C to Y-4 
proceed with actual readings from RAD Ops on scene

0923 Blue
Sr. Controller advised Blue-1 to inject message as appropriate to the S/S to go to a SAE for the 
event

0923 Yellow Yellow-1 reporting Fire Dept has reached 2 victims and preparing to extricate
0924 Blue ERO page moving response to level above ALERT
0924 Blue Based on analysis performed by S/S, classification of emergency has been moved to SAE
0925 Blue Blue-4 reporting patient #5 is being transported to University Hospital (simulated) nby RFFD

0925 Blue
Med1-1, 
Med 1-2 Message #Med1-1 & Med 1-2 dropped by Blue 4

0925 Green
was informed that 50 people walked through the plume and reported 1000 dpm of contamination 
on shoes, etc.

0925 Red
Info:  Control Cell Observation by Ops Mgr - LS/DW announcement is incorrect.  Emergency @ 
ALERT, however, we are sheltering on Site

0925 White
Additional TCP at SW371 gate.  240-NW of building is in respirator waiting for RCT  to  
monitor him

0927 Blue Blue-1 reporting ICP received info that there are 4 RED patients

0927 Red

PA announcement @ 0925 was prepared by PI team, approved by PI Mgr.  At time 
announcement was prepared, Site was at Alert status.  Sitewide shelter order was questioned by 
ED

0928 Green informed that RFST rec’d wind & weather info from HAC that is different from Control Cell
0929 Yellow Yellow-2 reporting 2 victims out of building & w/the Fire Dept
0930 Green reported current weather to green net
0930 White CAS announcement - EOC has ordered sheltering
0931 Blue IC providing a briefing to the CM on events as he understands them (IC = S/S)
0931 Red Crisis Mgr assumes overall responsibility for sitewide emergency response operations
0931 Red per red-1, EOC staffed and declared operational @ 0931
0931 Yellow Yellow-2 reporting preliminary search of facility completed
0932 Blue patient #1 is being transported from scene to University Hospital by RFFD Med #1
0932 Red Weather update
0932 Red confirmation that upgraded to SAE @ 0932, METS at 0936
0932 White LIMA2 allow ambulance to exit portal 2
0933 Red 2nd offsite notifications were completed at 0933
0934 Blue Blue-4 reporting patient #1 in RFFD Med 1 being transported to St. Anthony’s North
0934 Blue Med 2-1 message Med 2-1 dropped
0934 Red per yellow-4, accountability was complete at 0934.  Red net informed

0934 Yellow
Yellow-4 reports  accountability of personnel completed @ 0934 - personnel that are not 
contaminated are moving to B771
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0935 Green
Green-1 reporting contamination on BEST members from B371 @ ICP approx 100 cpm on 
booties and legs

0935 White 241 sent to keep evacuated employees off the road and in one area

0935 Yellow

Yellow-2 reporting fire team in building state protocol is to remain which will negate forced re-
entry and search - E/D dynamic injection message - tell Fire Dept they are needed at ICP - exit 
the facility

0936 Blue
Blue-7 advised that the dynamic message inject regarding secondary search & reentry was 
dropped

0936 Blue Blue-4 questioned if St. Anthony’s North is actually on divert

0937 Blue
ED advised Blue-1 @ ICP that a dynamic message inject was being made to force Fire Dept to 
exit the building to prepare for re-entry

0937 Yellow Yellow-2 reporting dynamic injection message dropped and Fire Dept exiting the facility
0938 Red Red-3 reporting St. Anthony’s North is not on "divert" status & is available to play

0939 Blue
the 5 victims have been identified to IC as missing; identified as missing as a result of the 
accountability process

0940 Blue told blue-4 that St Anthony’s North is NOT on divert & they are participating in the exercise

0940 Green Green-3 reported that 2 injured/contaminated "victims" transported to OMD have been contained
0940 Green Green-7 reports RFST ALPHA is at sampling point 9B and report 10 cpm on airsample
0942 Red 6C Message #6C was dropped
0942 White Last Fire Dept unit in thru Portal 2

0943 Blue
blue-4 reporting RFFD Med 2 enroute to T130E to drop off patient #4 and then will be enroute to 
St Anthony’s North w/patient #1

0944 JIC Media Advisory #1
0944 White Fire Dept unit went east - stopped by SPO’s at TCP - sent back west
0945 White 2nd ambulance out thru Portal 2

0949 Yellow
asked Yellow-4 the status of accountability.  BEST members tags pulled and 9 personnel/tags not 
yet accounted for

0950 Green was told that green-5 was going to the N sector of B374 to control survey activities by RCT’s
0950 Green green-7 reports that RAD FST ALPHA detects 250 cpm on areawipe @ sampling point 9B
0952 Yellow Fire Dept is standing on exterior stairwell and preparing for re-entry

0954 Blue
Blue-1 requested that issue with security for offsite transport of patient in Med 2; advised bue-1 
issue is resolved

0954 Green
Green-1 reports @ ICP that 3 BEST team members were deconned and isolated @ ICP boundary.  
Putting on white clothing-waiting further instruction

0954 Received via fax to 5206 Emergency Offsite Notification Form-update
0955 Blue IC conducting an incident briefing with ICP staff
0955 Blue RFFD Med #1 & Med #2 dropped one patient each @ University Hosp (simulated at T130E)
0955 White confusion over which unit (ambulance) is allowed to leave Site.
0956 Yellow told yellow-4 accountability completed yellow-4 to move to ICP - NE corner of B374

0958 Blue issue continued re: offsite patient transport; Med unit confusion being resolved w/LIMA units
1000 Green contacted Green-9 to report LS/DW update to the Site Emergency situation
1000 Yellow yellow 1 reports Fire Dept doffing bunker gear - going down due to heat
1003 Blue Blue-4 advised that Med 2 is transporting patient #2 to University Hospital (simulated T130E)
1003 Green Green-7 reports that FST Alpha detects 175 cpm on large area wipe at location 7B
1004 Blue Broomfield Ambulance has arrived at the incident scene

1005 Green
Green-7 reports that FST ALPHA is @ sampling point 6B and has reported background on large 
area wipe

1005 Green Green-8 contact green-7 for actual background counts from FST Alpha survey data
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1005 Green
Green-3 reporting HAZMAT individuals were requesting exemption from exercise.  Told green-3 
to check with IC at scene to confirm

1005 Red
info: crisis manager has directed B707 DO NOT access any more nuclear material & perform no 
add’l nuclear material movements until further notice

1009 Green Green-7 reports FST Alpha report background levels on large area wipes @ sampling point 5B
1010 Blue Blue-1 advised that ICP/IC has all B371/374 personnel accounted for except victim #3

1010 Green
Green-3 reporting that HAZMAT members in B374 are exempt from play unitl re-entry is made 
into B374

1010 Yellow
asking yellow-1 if Fire Dept made second facility entry.  Yellow-1 reporting "no".  Currently @ 
ICP discussing current building conditions with IC

1011 Blue Blue-1 asked from ICP if Building alarms could be silenced
1012 Blue victim #2 has arrived @ University Hosp (T130E)
1015 Blue IC conducting and ICP briefing
1015 Green Green-9 reports no victims transported to OMD decon as of 1015.  Will be advised.

1017 Green
Green-8 reports FST Bravo @ intersection of hwy 93 & hwy 72 conducting air sampling .  
W/report data to HAC

1017 Yellow ICP aware of missing person - Fire Dept organizing for re-entry and subsequent facility search

1018 Red
red-2 informed control cell that Dick Raaz has stated that "B771 has exceeded occupancy limits".  
Need direction from ED.  Per ED play out this issue.

1022 Blue LS/DW requesting Virginia Mackes contact X2446

1022 Red
Env. FWC completed (simulation) of diverting water treatment influence to influence storage 
tanks

1022 Red Jim Flourke was appointed as Recovery Manager @ 1022
1023 Blue 9 message #9 dropped
1024 Blue LS/DW announcement heard @ T130E updating personnel on exercise incident status
1025 Blue 11C message #11C dropped
1025 Green Provided update to B122 & B125 on Site emergency that was provided by LS/DW

1026 Blue
IC completing re-entry planning for FD; re-entry in approx 5 minutes to locate & remove victim 
#3

1029 Blue 4 FFS entering (re-entering) for search and rescue of victim #3
1029 Blue IC conducting ICP briefing on reentry operations and FD search & rescue
1029 Red Request per ED has Recovery Manager been appointed?  Per Red-1, negative
1029 Yellow Prepare for Fire Dept emergency re-entry

1030 Green
Green-2 reports 4 Fire Dept personnel are attempting re-entry into B374 to locate unaccounted 
for victim

1032 Red No information systems advisor has reported to the CSS
1033 Green Green-2 reports 3 Fire Dept personnel that will re-enter B374 to conduct search

1034 Blue
Blue-1 expressed safety concerns with the delay in getting drinking water to the scene.  ED 
working this issue with Red control units

1034 Yellow
Yellow 2 to Y Control two fire fighters @ door #1 preparing for re-entry and search for missing 
person

1035 Green
Was informed by Green-1 that 41 individuals suspected of being contaminated are being 
transported to B771 for decontamination and a controller is needed

1035 Red
CM has directed some B771 occupants to leave building (against sheltering) in order to resolve 
the occupancy issue (overcrowding)

1035 Red
Red-2 has injected additional information re: high FST rad data to alleviate confusion in HAC 
and EOC

1035 White Unit 240 (TCP) being removed from scene and transported to 771 for decon
1037 Yellow Fire Dept realizes no keys for door #11 searching for keys/door support

1038 Green
Green-8 reported FST Bravo is @ Sampling Point 62 conducting air sampling and reported 
building counts
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1041 Blue From B7, Patient #3 contact made and message #11 dropped on scene
1041 Yellow Yellow-2 to Yellow-Control Fire Dept has made reentry into B374
1042 Blue Victim #3 out of the building, being moved to hotline for medical evaluation

1042 Blue
IC received info that patient #3 has been found and is being removed by the FD, IC declared 
accountability complete

1043 Yellow Yellow-2 to Yellow-Control patient has been located in Room 3813 of B374
1044 Received via fax to 5206 Emergency Offsite Notification Form-update
1045 Green Green-9 requested status of 3rd victim from B374 enroute to OMD Decon Trailer
1045 Yellow Yellow-1 to Yellow-Control last victim has been removed from Room 3813 of B374
1046 Blue Med 4-1 message dropped

1048 Green
FST Alpha is on standby @ samplling point 5B recounting air sample filters.  Waiting further 
direction

1048 Green
Green-8 reports that FST Bravo is @ sampling point 93 and conducting ground surveys and air 
sampling

1050 Green
Green-1 informed G-Control that the 41 individuals sent to Bldg. 771 were deconned and 
released 20 mnutes ago

1050 Yellow Yellow-1 to Yellow-Control Broomfield Ambulance in place and ready for transport of victim
1051 Red Recovery Manager has arrived @ EOC

1054 Green
Was informed that FST Bravo Control will be turned over to the IC.  Team may go into the PA if 
requested

1055 Red ED requested that Red-1 call X5832

1055 Yellow
Yellow-control to Yellow-1 dynamic inject per E/O Manager - request facility management shut 
off building fire bells

1057 Blue
Advised Blue-1 and Blue-2 that IC needs to work through crowd issue in B750 and that personnel 
are not to be released to reenter 371/374

1058 Blue Broomfield Ambulance leaving scene with patient #3 and are enroute to OMD
1100 Green Was informed by Green-9 that patient #3 was enroute to OMD Decon in Bldg. 122

1100 Green
Was informed that the ICP at scene is being relocated to the NW area because of contamination at 
current location

1100 Green Was informated by Green-9 that 3rd victim has arrived @ OMD Decon

1100 Red
Red-1 will inject message to WSI Rep in EOC to make site announcecment for plant population 
"Do Not Call CAS or SAS to request B374 alarms be turned off"

1100 Yellow Yellow-1 to Yellow-Control victim #3 (last) with Broomfield Ambulance enroute to OMD
1101 Blue Message 5-1 - Blue-4 advised message has been dropped

1102 Red
Red-3 reported to control cell actual med response.  50 yr old male fall victim w/hand injury. 
Transporting actual injured patient to OMD via portal 2

1103 Blue Broomfield Ambulance arrived at OMD with patient #3

1103 Green
Green-9 reports actual injured patient being sent to OMD.  Gave instruction to continue play as 
advised

1104 Blue
ICP is moving to NW corner of B 371/374 due to rad readings of 100-200 cpm at previous 
location; route of travel west along north road

1105 Blue
Blue-11 informed control that OMD has been notified that the Fire Dept is transporting an 
ACTUAL fall victim to OMD; this does not impact their exercise participation

1105 Green
Green-4 reported 3rd victim was locatead and being transported to OMD with 240 dpm on face, 
less than MPA on stretcher and containment 

1108 Green
Was informed that FST alpha vehicle was obtaining gasoline because external dosimetry didn’t 
fill up after use

1110 Green PAC II & HAC requests RAD Con controller to meet FST at PACII.  Green-1 w/follow up

1113 Red

Red-3 reporting Fire officers on scene requesting approval from ED to re-enter B374 to reset fire 
phones and turn off fire alarms.  Per ED that request must go thru EOC (re-entry function).  Info 
passed to red-3 - forward request thru EOC
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1115 Blue
Weather update delivered to Blue Net - 66 degrees F, 334 degrees NW to SE, 32% humidity, 8 
mph - gust to 10 mph

1115 Blue
IC delivering briefing to ICP staff in preparation to turn command over to bldg mgmt.  Working 
on closing out emergency operations

1115 Green
Team Bravo is at Gas house awaiting fuel.  Has been given direction to report to PAC 2 after 
fueling

1116 JIC Media Advisory #2

1116 Yellow
Yellow-Control to Yellow Net weather update - 66 degrees F, 334 degrees NW to SE, 32% 
humidity, 8 mph - gust to 10 mph

1120 Green
Green-7 reports FS Team alpha @ sapling point 8B conducting ground disposition surveys as 
directed

1121 Blue
Blue-11 advised that patient #3 in decon trailer receiving treatment (actual patient -fall victim- 
also being treated by non-exercise staff)

1121 Red
red-3 reporting status of actual response:  victim has lacerated elbow and is currently being 
treated at OMD

1122 Yellow Yellow-2 reports Fire Dept requesting to start cleanup of hoses - approved per  ED

1124 Green
Green-1 reports to PACII as controller was informed it has been locked down.  As part of play, 
sheltering by site

1126 Red
Env. FWC requesting to send engineer to pull the effluent filter from B374 stack.  ED did not 
approve, this is part of recovery planning.  Red-2 should inject this info at this time

1129 Green
Green-7 reported that FST Alpha was directed to conduct surveys at sampling points onsite.  Will 
advise points A & Q

1130 Red
Decision made to lift Site sheltering except for immediate area.  Announcement to be made 
shortly regarding lifting shelter & extended lunch hours.

1132 Blue
Patient #3 released by OMD; OMD personnel currently working on evaluating and cleaning 
OMD decon trailer

1133 Green Green-1 is @ the  NW corner of B374 watching activities of Fire Dept & RCTs.  Is not @ PAC II
1135 Green Green-7 reports that 37 cpm was detected @ sampling point ALPHA by FST ALPHA

1140 Green
Green-9 reports 3rd victim was deconned & treated at OMD & transferred to offsite hospital 
(simulated)

1140 Green Green-7 reports FST Bravo reporting 30 cpm @ sampling point Q
1144 Blue Blue-2 reports drinking water arrived at the relocated ICP at approx 1100
1145 Green reporting to green net sheltering has been lifted
1145 Green field play terminated-conduct hot wash & attend C/E debrief in T130E

1145 Red
Requested EOC call Bill @ X4334 to extend lunch hours to 1:30 per ED.  Per Red-1, Nancy Tour 
& Bob Card said 1:00 is sufficient

1147 CC Termiate Field Play
1157 CC EOC Terminated


